Author's response to reviews

Title: Survey of oxaliplatin-associated neurotoxicity using an interview-based questionnaire in patients with metastatic colorectal cancer

Authors:

Gregory D Leonard (LeonardG@Sehb.ie)
Maurice A Wright (Wrightwils@aol.com)
Mary G Quinn (mquinn@tech-res.com)
Suzanne Fioravanti (fioravas@mail.nih.gov)
Nancy Harold (nharold@mail.nih.gov)
Barbara Schuler (schulerb@mail.nih.gov)
Rebecca R Thomas (rebeccathomasmmd@yahoo.com)
Jean L Grem (jgrem@unmc.edu)

Version: 5 Date: 28 August 2005

Author's response to reviews:

1. Sentence Case is now used for all section headings and sub-headings in the manuscript.

2. "Author's contributions" has now been changed to "Authors' contributions".

3. the Authors' contribution statements have been revised as suggested to conform to biomedcentral.com format.